Author's response to reviews

Title: Complete response in gallbladder cancer to Erlotinib plus Gemcitabine does not require mutation of the epidermal growth factor receptor gene: a case report

Authors:

Kabir Mody (kabir.mody@norwalkhealth.org)
Edward Strauss (edward.strauss@norwalkhealth.org)
Robert Lincer (robert.lincer@norwalkhealth.org)
Richard C Frank (richard.frank@norwalkhealth.org)

Version: 5 Date: 18 October 2010

Author's response to reviews: see over
October 18, 2010

Dear BMC Cancer Editors:

Re: MS: 2132589838408653

The following final changes have been made to our submission to BMC Cancer entitled, “Complete response in gallbladder cancer to Erlotinib plus Gemcitabine does not require mutation of the epidermal growth factor receptor gene: a case report,” by K. Mody, et al.

1. Authors’ contribution now clearly indicates that all authors read and approved the final manuscript.
2. References have been corrected such that they are now in numerical order in the manuscript.

Thank you so much for your consideration of our manuscript.

Respectfully submitted,

Richard C. Frank, MD
Norwalk Hospital
24 Stevens Street
Norwalk, CT 06856
Richard.frank@norwalkhealth.org
Phone: 203-845-4896
Fax: 203-855-3539